Author's response to reviews

Title: Characterization of the Inflammatory Cell Infiltrate and Expression of Costimulatory Molecules in Chronic Echinococcus Granulosus Infection of the Human Liver

Authors:

Ali Vatankhah (avatankhah@hotmail.com)
Judit Halász (halaszjudi@gmail.com)
Violetta Piurko (piurko.violetta@gmail.com)
Tamás Barbai (tbarbai@gmail.com)
Erzsébet Rásó (rasoerzs@gmail.com)
Prof Jozsef Timar (jitimar@gmail.com)

Version: 5  Date: 1 October 2015

Author's response to reviews:

1/ We have authenticated the abstract in the online system and the manuscript body.
2/ On lane 128/129 we declare that we actually had an IRB approval for this retrospective study. (TUKEB15/2008).
3/ We indicate now the Authors contribution and their approval (lane 17) and corrected their initials.
4/ We have included Tables 1-3 into the text of the manuscript lane 640-671.
5/ We have corrected the English highlighted by yellow.